Author's response to reviews

**Title:** Phase II study of preoperative radiation plus concurrent daily tegafur-uracil (UFT) with leucovorin for locally advanced rectal cancer

**Authors:**

Patrice Cellier (p.cellier@angers.fnclcc.fr)
Bernard Leduc (oncologie@ch-brive.fr)
Laurent Martin (onco.lehavre@wanadoo.fr)
Brigitte Vié (b.vie@baclesse.fr)
Christian Chevelle (c.chevelle@clinique-pasteur.com)
Véronique Vendrely (veronique.vendrely@chu-bordeaux.fr)
Augustin Salemcourt (docsalemkour@yahoo.fr)
Christian Carrie (carrie@lyon.fnclcc.fr)
Gilles Calais (calais@med.univ-tours.fr)
Pascal Burtin (pascal.burtin@igr.fr)
Loïc Campion (l-campion@nantes.fnclcc.fr)
Michèle Boisdrón-Celle (m.boisdron@angers.fnclcc.fr)
Alain Morel (a.morel@angers.fnclcc.fr)
Virginie Berger (v.berger@angers.fnclcc.fr)
Erick Gamelin (egamelin@amgen.com)

**Version:** 5  **Date:** 21 February 2011

**Author's response to reviews:**

Dear Editor,

This is the latest version of the manuscript entitled “Phase II study of preoperative radiation plus concurrent daily tegafur–uracil (UFT) with Leucovorin for locally advanced rectal cancer” by Patrice Cellier et al.

Author list was modified as asked in the mail from 16 of february.

References 44 and 45 were added in the text (p15) and the numerical order of others reference have been checked and modified.

Looking forward to reading this paper on line

Tours sincerely

For Michèle Boisdrón-Celle, Virginie BERGER